Author’s response to reviews

Title: South African Primary Health Care Allied Health Clinical Practice Guidelines: The big picture

Authors:

Karen Grimmer (Karen.Grimmer@unisa.edu.au)
Shingai Machingaidze (Shingai.Machingaidze@mrc.ac.za)
Janine Dizon (janine.dizon@unisa.edu.au)
Quinette Louw (qalouw@sun.ac.za)
Hameedah Parker (hparker@sun.ac.za)

Version: 2 Date: 06 Jun 2017

Author’s response to reviews:

Responses attached as a separate document